Author's response to reviews

Title: Early detection of breast cancer: benefits and risks of supplemental breast ultrasound in asymptomatic women with mammographically dense breast tissue. A systematic review

Authors:

Monika Nothacker (nothacker@azq.de)
Volker Duda (duda@med.uni-marburg.de)
Markus Hahn (agmimi@joho.de)
Mathias Warm (Mathias.warm@medizin.uni-koeln.de)
Friedrich Degenhardt (frideg@t-online.de)
Helmut Madjar (Madjar.gyn@dkd-wiesbaden.de)
Susanne Weinbrenner (weinbrenner@azq.de)
Ute-Susann Albert (albertu@med.uni-marburg.de)

Version: 8 Date: 16 September 2009

Author's response to reviews:

Dear Production Team,
we made the following correction
Table 2:
Kaplan et al., deleted: et al.
Table 3:
Kaplan et al., deleted: et al.

With best regards
Ute-Susann Albert